Author’s response to reviews

Title: Valproic acid-induced amphiregulin secretion confers resistance to temozolomide treatment in human glioma cells

Authors:

Jui-Chieh Chen (jcc@mail.ncyu.edu.tw)
I-Neng Lee (ineng1105@cgmh.org.tw)
Cheng Huang (chengh@ym.edu.tw)
Yu-Ping Wu (ypw0601@cgmh.org.tw)
Chiu-Yen Chung (yen16@adm.cgmh.org.tw)
Ming-Hsueh Lee (ma2072@cgmh.org.tw)
Martin Hsiu-Chu Lin (m7099@cgmh.org.tw)
Jen-Tsung Yang (jents716@icloud.com)

Version: 4 Date: 02 Jun 2019

Author’s response to reviews:

Editor Comments:

1. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
Ans: According to the editor's suggestion, the author's name has been represented by the full initials, and the author's initials clearly represent the different authors. (Page17, line1)

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Ans: We have added the statement as follows: The funding bodies had no role in: the design of the study; data collection, analysis, or interpretation; or, in writing the manuscript. (Page16, line36)

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Ans: We now upload the final version of our manuscript and related documents in accordance with the above specifications.